Author's response to reviews

Title: Adoption and sustainability of decision support for patients facing health decisions: An implementation case study in nursing

Authors:

Dawn Stacey (dstacey@uottawa.ca)
Marie-Pascale Pomey (marie-pascale.pomey@umontreal.ca)
Annette M O'Connor (aoconnor@ohri.ca)
Ian D Graham (igraham@ohri.ca)

Version: 3 Date: 17 August 2006

Author's response to reviews: see over
ATTN:  Dawn Stacey  
Doctoral Candidate  
CIHR Doctoral Studies Award  
Population Health PhD Program  
University of Ottawa

PERMISSION: For use in PhD paper-based dissertation being submitted to the University of Ottawa on January 14th, called “Design and evaluation of an implementation strategy to enhance decision support by call centre nurses for callers facing values-sensitive health decisions,” as well as part of an article in a peer reviewed journal. Proper referencing of CJNR as the source for the material must be included in the published book.


TOTAL: $35.00

PAYMENT INFORMATION:
Enclosed is my cheque for $ __________ payable to Canadian Journal of Nursing Research

MasterCard #___________________________________ Expiry date_________________

Visa #_________________________________________ Expiry date_________________

Signature: _______________________________